#### REQUEST FOR PROPOSAL

# INTEGRATED HEADQUARTERS OF MINISTRY OF DEFENCE (NAVY) INDIAN NAVY SPORTS CONTROL BOARD IHQ OF MOD (NAVY) 7<sup>TH</sup> FLOOR, CHANAKYA BHAWAN CHANAKYA PURI, NEW DELHI – 110 021

# INVITATION OF QUOTATIONS FOR PROCUREMENT OF SPORTS KIT AND GEAR FOR THE NAVY TEAMS

Request For Proposal (RFP) No.

INSCB/1098/Kit & Gear/19-20 Dated May 19

- 1. Bids under two bids system (**Technical and Commercial**) in <u>separate sealed</u> covers are invited for supply of Kit Gear for Navy teams.
- 2. Please super scribe the above mentioned Title, RFP number and date of opening of the Bids on the sealed cover to avoid the Bid being declared invalid.
- 3. The address and contact numbers for sending Bids or seeking clarifications regarding this RFP are given below:-
  - (a) Bids / queries to be addressed to:-

The President
Indian Navy Sports Control Board
Integrated Headquarters of
Ministry of Defence (Navy)
7<sup>th</sup> Floor, Chanakya Bhawan
Chanakya Puri, New Delhi – 110 021

- (b) Postal address for sending the Bids: Same as 3 (a)
- (c) Name / designation of the contact personnel:
  - (i) Commodore Paresh Sawhney, President INSCB
  - (ii) Captain AK Bahl, Secretary INSCB
- (d) Telephone numbers of the contact personnel:
  - (i) 011-24674063
  - (ii) 011- 26887485
- (e) e-mail id of contact personnel: <a href="mailto:inscb-navy@nic.in">inscb-navy@nic.in</a>
- (f) Fax number: 011-26887485

- 4. This RFP is divided into five parts as follows:-
  - (a) Part I Contains General Information and instructions for the Bidders about the RFP such as the time, place of submission and opening of tenders, Validity period of tenders, etc
  - (b) Part II Contains essential details of the items/services required, such as the schedule of requirements (SOR), Technical specifications, Delivery Period, Mode of Delivery and Consignee details.
  - (c) Part III Contains Standard Conditions of RFP, which will form part of the contract with the successful Bidder.
  - (d) Part IV Contains Special Conditions applicable to this RFP and which will also form part of the contract with the successful Bidder.
  - (e) Part V Contains Evaluation Criteria and Format for Price Bids
- 5. This RFP is being used with no financial commitment and the Buyer reserves the right to change or vary and part thereof at any stage. Buyer also reserves the right to withdraw the RFP, should it become necessary at any stage.

(Paresh Sawhney)

Commodore

President INSCB (AOL)

2-8 May 19

#### Part I - General information

- 1. <u>Last date and time for depositing the Bids.</u> Your sealed quotation (Technical and Commercial) should reach this office latest by 1330 hrs on 25 Jun 19. The responsibility to ensure this lies with the bidder.
- 2. <u>Manner of Depositing the Bids.</u> Sealed Bids should be either dropped in the Tender Box marked as "<u>TENDER BOX SPORTS KIT & GEAR</u>" or sent by registered post at the address given above so as to reach by the due date and time. Late tenders will not be considered. No responsibility will be taken for postal delay or non-delivery / non-receipt of Bid documents. Bids sent by FAX or e-mail will not be considered.
- 3. <u>Time and date for opening of Bids.</u> 1400 hrs on 25 Jun 19 (If due to any exigency, the due date for opening of the Bids is declared a closed holiday, the Bids will be opened on the next working day at the same time or on any other day/time, as intimated). The bidders are requested to forward brochures of the proposed items along with the technical bids.
- 4. Location of the Tender Box. The sealed quotations addressed to The President, Indian Navy Sports Control Board, IHQ of MoD (Navy), 7<sup>th</sup> Floor Chanakya Bhawan, Chanakya Puri, New Delhi 110 021 should be dropped in Tender Box marked as "TENDER BOX SPORTS KIT & GEAR " kept near the reception office at INSCB. Only those Bids that are found in the tender box will be opened. Bids dropped in the wrong Tender Box will be rendered invalid.
- 5. Place of opening of the Bids. Office of the Secretary, Indian Navy Sports Control Board, IHQ of MoD (Navy), 7<sup>th</sup> Floor Chanakya Bhawan, Chanakya Puri, New Delhi-21. The Bidders may depute their representatives, duly authorized in writing, to attend the opening of Bids on the due date and time. Rates and important commercial / technical clauses quoted by all Bidders will be read out in the presence of the representatives of all the Bidders. This event will not be postponed due to non-presence of your representative.
- 6. <u>Two-Bid system.</u> Two-bid system will be followed and Technical Bid would be opened on the time and date mentioned above. Date of opening of the Commercial Bid will be intimated after acceptance of the Technical Bids. Commercial Bids of only those firms will be opened, whose Technical Bids are found compliant/suitable after Technical evaluation is done by the Buyer.
- 7. <u>Pre-Qualification Criteria</u>. Pre-qualification Criteria for opening of Technical and Commercial bids are as follows:-
  - (a) Firms must attend all the meetings including Pre bid Meeting to be attended by the firm's owner or representative of the firm's with the proper authorization to attend the meeting.
  - (b) The scrutiny of tender documents will be done by the Technical Evaluation Committee (TEC) constituted by the CFA on the first day of tender opening.
  - (c) Sample Evaluation will be done by the TEC for the **next two days** 27 Jun & 28 Jun 19 (first day all team items second day all common items) of tender opening and only firms would be entertained for sample valuation those who are found correct in all respect as mentioned above as per RFP.

- (d) Fabric Material testing can be undertaken at Government approved test laboratory, if considered necessary. The cost of the testing will need to be borne by the vendor.
- (e) This Organisation has the right to reject the tender of any firm who has been blacklisted by any government agency over the past three years.
- (f) Field trails may be undertaken for shoes for which vendor will have to supply entire pair.
- (g) This Organisation has the right to reject the tender of any firm who has not able to produce Genuineness Certificate as and when requested by technical evaluation committee (TEC).
- 8. Forwarding of Bids. Bids should be forwarded by Bidders under their original memo / letter pad inter alia furnishing details like TIN number, VAT/CST number, Bank address with EFT Account if applicable, etc and complete postal & e-mail address of their office.
- 9. <u>Clarification regarding contents of the RFP.</u> A prospective bidder who requires clarification regarding the contents of the bidding documents shall notify to the Buyer in writing about the clarifications sought not later than **14** (fourteen) days prior to the date of opening of the Bids. Copies of the query and clarification by the purchaser will be sent to all prospective bidders who have received the bidding documents.
- 10. <u>Modification and Withdrawal of Bids.</u> A bidder may modify or withdraw his bid after submission provided that the written notice of modification or withdrawal is received by the Buyer prior to deadline prescribed for submission of bids. A withdrawal notice may be sent by fax but it should be followed by a signed confirmation copy to be sent by post and such signed confirmation should reach the purchaser not later than the deadline for submission of bids. No bid shall be modified after the deadline for submission of bids. No bid may be withdrawn in the interval between the deadline for submission of bids and expiration of the period of bid validity specified. Withdrawal of a bid during this period will result in Bidder's forfeiture of bid security.
- 11. Clarification regarding contents of the Bids. During evaluation and comparison of bids, the Buyer may, at its discretion, ask the bidder for clarification of his bid. The request for clarification will be given in writing and no change in prices or substance of the bid will be sought, offered or permitted. No post-bid clarification on the initiative of the bidder will be entertained.
- 12. Rejection of Bids. Canvassing by the Bidder in any form, unsolicited letter and post-tender correction may invoke summary rejection with forfeiture of EMD. Conditional tenders will be rejected.
- 13. <u>Unwillingness to quote.</u> Bidders unwilling to quote should ensure that intimation to this effect reaches before the due date and time of opening of the Bid, failing which the defaulting Bidder may be de-listed for the given range of items as mentioned in this RFP.
- 14. <u>Validity of Bids.</u> The Bids should remain valid for **180 days from the last date** of submission of the Bids.

Earnest Money Deposit. Bidders are required to submit Earnest Money Deposit (EMD) minimum Rs 10000/- or 3% of quoted items along with their bids. The EMD may be submitted in favor of the Secretary, Indian Navy Sports Control Board in the form of an Account Payee Demand Draft, Fixed Deposit Receipt, Banker's Cheque or Bank Guarantee from any of the public sector banks or a private sector bank authorized to conduct government business as per Form DPM-16 (Available in MoD website and can be provided on request). EMD is to remain valid for a period of forty-five days beyond the final bid validity period. EMD of the unsuccessful bidders will be returned to them at the earliest after expiry of the final bid validity and latest on or before the 30th day after the award of the contract. The Bid Security of the successful bidder would be returned, without any interest whatsoever, after the receipt of Performance Security from them as called for in the contract. EMD is not required to be submitted by those Bidders who are registered with the Central Purchase Organization (e.g. DGS&D), National Small Industries Corporation (NSIC) or any Department of MoD or MoD itself only for those items registered items with NSIC. In case a firm quotes for NSIC items as well as other items then EMD will have to be paid for an amount equal to minimum Rs 10000/- or 3% of the value of Non NSIC items quotes by the firm. The EMD will be forfeited if the bidder withdraws or amends impairs or derogates from the tender in any respect within the validity period of their tender.

> (Paresh Sawhney) Commodore

President INSCB (AOL)

#### Part II - Essential Details of Items/Services required

- 1. <u>Schedule of Requirements</u> As per list placed at Appendix 'A' to Tender INSCB/1098/Kit & Gear/19-20 Dated May 19.
- 2. <u>Technical Details</u> The information in the technical bid should be complete in itself to facilitate full technical scrutiny. The vendor must volunteer all information required for this purpose. The Navy reserves the right to seek necessary clarifications on the technical bids. It must be borne in mind that no change to commercial bid, arising out of clarifications on technical bid, is permitted. The Article number / necessary technical details of the items required are also contained in Appendix 'A' to this RFP.
- 3. <u>Two-Bid System</u> In respect of Two-bid system, Bidders are required to furnish clause by clause compliance of specifications bringing out clearly the deviations from specification, if any. All the deviations are to be mentioned in the matrix below only. Any deviations mentioned anywhere else will be disregarded. The Bidders are to submit the compliance statement on company's letter head with signature and stamp in the following format only along with Technical Bid —

<u>SI</u>	Para of RFP/SOR specifications item-wise	Specification of item offered as per RFP/SOR	Compliance to RFP specification  - whether Yes / No	In case of non compliance, deviation from RFP to be specified in unambiguous terms
-----------	--	--	---	--

- 4. <u>Delivery Period</u> Delivery period for supply of items at the ultimate consignee would be <u>90 days</u> from the effective date of contract. Please note that Contract can be cancelled unilaterally by the Buyer in case items are not received within the contracted delivery period. Extension of contracted delivery period will be at the sole discretion of the Buyer, with applicability of LD clause.
- Ultimate Consignee details
  - (a) The Secretary, INSCB for Delhi Based team
  - (b) The Joint Secretaries of INSCC (MB), INSCC (V), INSCC(K) Oi/CS Navy teams.

(Paresh Sawhney)

Commodore

President INSCB (AOL)

## Part III – Standard Conditions of RFP

The Bidder is required to give confirmation of their acceptance of the Standard Conditions of the Request for Proposal mentioned below which will automatically be considered as part of the Contract concluded with the successful Bidder (i.e. Seller in the Contract) as selected by the Buyer. Failure to do so may result in rejection of the Bid submitted by the Bidder.

- 1. <u>Law</u>. The Contract shall be considered and made in accordance with the laws of the Republic of India. The contract shall be governed by and interpreted in accordance with the laws of the Republic of India.
- 2. <u>Effective Date of the Contract</u>. The contract shall come into effect on the date of signatures of both the parties on the contract (Effective Date) and shall remain valid until the completion of the obligations of the parties under the contract. The deliveries and supplies and performance of the services shall commence from the effective date of the contract.
- 3. Arbitration. All disputes or differences arising out of or in connection with the Contract shall be settled by bilateral discussions. Any dispute, disagreement or question arising out of or relating to the Contract or relating to construction or performance, which cannot be settled amicably, may be resolved through arbitration. The standard clause of arbitration is as per Forms DPM-7, DPM-8 and DPM-9 (Available in MoD website and can be provided on request).
- Penalty for use of Undue influence. The Seller undertakes that he has not given, offered or promised to give, directly or indirectly, any gift, consideration, reward, commission, fees, brokerage or inducement to any person in service of the Buyer or otherwise in procuring the Contracts or forbearing to do or for having done or forborne to do any act in relation to the obtaining or execution of the present Contract or any other Contract with the Government of India for showing or forbearing to show favour or disfavour to any person in relation to the present Contract or any other Contract with the Government of India. Any breach of the aforesaid undertaking by the Seller or any one employed by him or acting on his behalf (whether with or without the knowledge of the Seller) or the commission of any offers by the Seller or anyone employed by him or acting on his behalf, as defined in Chapter IX of the Indian Penal Code, 1860 or the Prevention of Corruption Act, 1986 or any other Act enacted for the prevention of corruption shall entitle the Buyer to cancel the contract and all or any other contracts with the Seller and recover from the Seller the amount of any loss arising from such cancellation. A decision of the Buyer or his nominee to the effect that a breach of the undertaking had been committed shall be final and binding on the Seller. Giving or offering of any gift, bribe or inducement or any attempt at any such act on behalf of the Seller towards any officer/employee of the Buyer or to any other person in a position to influence any officer/employee of the Buyer for showing any favour in relation to this or any other contract, shall render the Seller to such liability/ penalty as the Buyer may deem proper, including but not limited to termination of the contract, imposition of penal damages, forfeiture of the Bank Guarantee and refund of the amounts paid by the Buyer.
- 5. <u>Agents / Agency Commission</u>. The Seller confirms and declares to the Buyer that the Seller is the original manufacturer of the stores/provider of the services referred to in this Contract and has not engaged any individual or firm, whether Indian or foreign

whatsoever, to intercede, facilitate or in any way to recommend to the Government d. India or any of its functionaries, whether officially or unofficially, to the award of the contract to the Seller; nor has any amount been paid, promised or intended to be paid to any such individual or firm in respect of any such intercession, facilitation or recommendation. The Seller agrees that if it is established at any time to the satisfaction of the Buyer that the present declaration is in any way incorrect or if at a later stage it is discovered by the Buyer that the Seller has engaged any such individual/firm, and paid or intended to pay any amount, gift, reward, fees, commission or consideration to such person, party, firm or institution, whether before or after the signing of this contract, the Seller will be liable to refund that amount to the Buyer. The Seller will also be debarred from entering into any supply Contract with the Government of India for a minimum period of five years. The Buyer will also have a right to consider cancellation of the Contract either wholly or in part, without any entitlement or compensation to the Seller who shall in such an event be liable to refund all payments made by the Buyer in terms of the Contract along with interest at the rate of 2% per annum above LIBOR rate. The Buyer will also have the right to recover any such amount from any contracts concluded earlier with the Government of India.

- 6. Access to Books of Accounts. In case it is found to the satisfaction of the Buyer that the Seller has engaged an Agent or paid commission or influenced any person to obtain the contract as described in clauses relating to Agents/Agency Commission and penalty for use of undue influence, the Seller, on a specific request of the Buyer, shall provide necessary information/ inspection of the relevant financial documents/information.
- 7. Non-Disclosure of Contract Documents. Except with the written consent of the Buyer/ Seller, other party shall not disclose the contract or any provision, specification, plan, design, pattern, sample or information thereof to any third party.
- 8. <u>Liquidated Damages</u>. In the event of the Seller's failure to submit the Bonds, Guarantees and Documents, required items, the Buyer may, at his discretion, withhold any payment until the completion of the contract. The BUYER may also deduct from the SELLER as agreed, liquidated damages to the sum of 0.5% of the contract price of the delayed/undelivered stores/services mentioned above for every week of delay or part of a week, subject to the maximum value of the Liquidated Damages being not higher than 10% of the value of delayed stores.
- 9. <u>Termination of Contract</u>. Buyer shall have the right to terminate this Contract in part or in full in any of the following cases:-
  - (a) The delivery of the material is delayed for causes not attributable to Force Majeure for more than (90 days) after the scheduled date of delivery.
  - (b) The Seller is declared bankrupt or becomes insolvent.
  - (c) The supply of items is delayed due to causes of Force Majeure by more than 06 months, provided Force Majeure clause is included in contract.
  - (d) The Buyer has noticed that the Seller has utilised the services of any Indian/Foreign agent in getting this contract and paid any commission to such individual/company etc.
  - (e) As per decision of the Arbitration Tribunal.

- 10. <u>Notices</u>. Any notice required or permitted by the contract shall be written in the English language and may be delivered personally or may be sent by FAX or registered pre-paid mail/airmail, addressed to the last known address of the party to whom it is sent.
- 11. <u>Transfer and Sub-letting</u>. The Seller has no right to give, bargain, sell, assign or sublet or otherwise dispose of the Contract or any part thereof, as well as to give or to let a third party take benefit or advantage of the present Contract or any part thereof.
- 12. Patents and other Industrial Property Rights. The prices stated in the present Contract shall be deemed to include all amounts payable for the use of patents, copyrights, registered charges, trademarks and payments for any other industrial property rights. The Seller shall indemnify the Buyer against all claims from a third party at any time on account of the infringement of any or all the rights mentioned in the previous paragraphs, whether such claims arise in respect of manufacture or use. The Seller shall be responsible for the completion of the supplies irrespective of the fact of infringement of the supplies, irrespective of the fact of infringement of any or all the rights mentioned above.
- 13. <u>Amendments</u>. No provision of present Contract shall be changed or modified in any way (including this provision) either in whole or in part except by an instrument in writing made after the date of this Contract and signed on behalf of both the parties and which expressly states to amend the present Contract.

#### 14. Taxes and Duties

#### (a) General.

- (i) Bidder must indicate separately the relevant taxes / Duties likely to be paid in connection with delivery of completed goods specified in RFP. In absence of this, the total cost quoted by them in their bids will be taken into account in the ranking of bids.
- (ii) If a Bidder is exempted from payment of any duty/tax up to any value of supplies from them, he should clearly state that no such duty/tax will be charged by him up to the limit of exemption which they may have. If any concession is available in regard to rate/quantum of any Duty/tax, it should be brought out clearly. In such case, relevant certificate will be issued by the Buyer to enable the Seller to obtain exemptions from taxation authorities.
- (iii) Any changes in levies, taxes and duties levied by Central / State / Local governments such as excise duty, VAT, Service tax, Octroi / entry tax, etc on final product upward as a result of any statutory variation taking place within contract period shall be allowed reimbursement by the Buyer, to the extent of actual quantum of such duty / tax paid by the Seller. Similarly, in case of downward revision in any such duty / tax, the actual quantum of reduction of such duty / tax shall be reimbursed to the buyer by the Seller. All such adjustments shall include all reliefs, exemptions rebates, concession etc, if any, obtained by the Seller. Section 64-A od Sales of Goods Act will be relevant in this situation.

- (iv) Levies, taxes and duties levied by Central / State / Local governments such as excise, VAT, Service tax, Octroi. Entry tax etc on final product will be paid by the Buyer on actuals, based on relevant documentary evidence. Taxes and duties on input items will not be paid by the Buyer and they may not be indicated separately in the bids. Bidders are required to include the same in the pricing of their product.
- (b) <u>Customs Duty</u>. Customs Duty Exemption Certificate (CDEC) from SSCB will be provided for the value of stores to be imported. In case of non-acceptance of the CDEC by the Customs Authorities, the seller shall initially pay the customs duty at the prescribed rate and subsequently claim Customs duty as actually paid as follows:-.

  - (ii) Subsequent to the reimbursement of customs duty, the Bidder will submit to the concerned Payment Authority a certificate to the effect that he has not obtained any refund of customs duty subsequent to the payment of duty to the Customs authority by him. In addition, he shall also submit to the Paying Authority concerned a certificate immediately after a period of three months from the date of payment of the duty to customs authorities to the effect that he has not applied for refund of the customs duty subsequent to the payment of duty to the customs authorities by him.
  - (iii) In case the Bidder obtains any refund of customs duty, subsequently to the payment of the same by him to the customs authorities and reimbursement of the customs duty to him by the Payment Authority, he should forthwith furnish the details of the refund obtained and afford full credit of the same to the Buyer.

#### (c) GST.

- (i) If it is desired by the Bidder to ask for GST to be paid as extra, the same must be specifically stated. In the absence of any such stipulation in the bid, it will be presumed that the prices quoted by the Bidder are inclusive of GST and no liability of GST will be developed upon the Buyer.
- (ii) On the Bids quoting GST extra, the rate and the nature of GST applicable at the time of supply should be shown separately along with HSN code. GST will be paid to the Seller at the rate at which it is liable to

be assessed or has actually been assessed provided the transaction of sale is legally liable to GST and the same is payable as per the terms of the contract.

(iii) The seller confirms that any decrease in the total costs due to change in the tax structure after implementation of GST or on account of input tax audit, during the period of the contract will be passed to the buyer.

(Paresh Sawhney)

Commodore

President INSCB (AOL)

#### Part IV - Special Conditions of RFP

The Bidder is required to give confirmation of their acceptance of Special Conditions of the RFP mentioned below which will automatically be considered as part of the Contract concluded with the successful Bidder (i.e. Seller in the Contract) as selected by the Buyer. Failure to do so may result in rejection of Bid submitted by the Bidder.

- 1. Performance Guarantee. The Bidder will be required to furnish a Performance Guarantee by way of Bank Guarantee in favor of the Secretary, Indian Navy Sports Control Board through a public sector bank or a private sector bank authorized to conduct government business (ICICI Bank Ltd., Axis Bank Ltd or HDFC Bank Ltd.) for a sum equal to 10% of the contract value within 30 days of receipt of the confirmed order. Performance Bank Guarantee should be valid up to 60 days beyond the date of warranty. The specimen of PBG is given in Form DPM-15 (Available in MoD website and can be provided on request).
- 2. Option Clause. The contract will have an Option Clause, wherein the Buyer can exercise an option to procure an additional 50% of the original contracted quantity in accordance with the same terms & conditions of the present contract. This will be applicable within the currency of contract. The Bidder is to confirm the acceptance of the same for inclusion in the contract. It will be entirely the discretion of the Buyer to exercise this option or not.
- 3. Repeat Order Clause. The contract will have a Repeat Order Clause, wherein the Buyer can order upto 50% quantity of the items under the present contract within six months from the date of supply/successful completion of this contract, the cost, terms & conditions remaining the same. The Bidder is to confirm acceptance of this clause. It will be entirely the discretion of the Buyer to place the Repeat order or not.
- 4. <u>Tolerance Clause</u>. To take care of any change in the requirement during the period starting from issue of RFP till placement of the contract, Buyer reserves the right to 50 % plus/ minus increase or decrease the quantity of the required goods up to that limit without any change in the terms & conditions and prices quoted by the Seller. While awarding the contract, the quantity ordered can be increased or decreased by the Buyer within this tolerance limit.
- 5. Payment Terms. It will be mandatory for the Bidders to indicate their bank account numbers and other relevant e-payment details so that payments could be made through ECS/EFT mechanism instead of payment through cheques, wherever feasible. A copy of the model mandate form prescribed by RBI to be submitted by Bidders for receiving payments through ECS is at Form DPM-11 (Available in MoD website and can be given on request). 100% Payment will be made on receipt of items in good condition by consignees along with inspection certificate issued by the end user and acceptance by the buyer.
- 6. Advance Payments. No advance payment(s) will be made.
- 7. Paying Authority. The CDA (New Delhi) R.K Puram. The payment of bills will be made on submission of the following documents by the seller to the paying Authority along with the bill:-

- (a) Ink-signed copy of contingent bill / Seller's bill.
- (b) Ink-signed copy of Commercial invoice / Seller's bill.
- (c) Copy of Supply Order/Contract with U.O. number and date of IFA's concurrence, where required under delegation of powers.
- (d) CRVs in duplicate.
- (e) Inspection note.
- (f) Claim for statutory and other levies to be supported with requisite documents / proof of payment such as Excise duty challan, Customs duty clearance certificate, Octroi receipt, proof of payment for EPF/ESIC contribution with nominal roll of beneficiaries, etc as applicable.
- (g) Exemption certificate for Excise duty / Customs duty, if applicable.
- (h) Bank guarantee for advance, if any.
- (j) Guarantee / Warranty certificate.
- (k) Performance Bank guarantee / Indemnity bond where applicable.
- (I) DP extension letter with CFA's sanction, U.O. number and date of IFA's concurrence, where required under delegation of powers, indicating whether extension is with or without LD.
- (m) Details for electronic payment viz Account holder's name, Bank name, Branch name and address, Account type, Account number, IFSC code, MICR code (if these details are not incorporated in supply order/contract).
- (n) Any other document / certificate that may be provided for in the Supply Order / Contract.
- (p) User Acceptance.
- (q) Xerox copy of PBG.

(Note – From the above list, the documents that may be required depending upon the peculiarities of the procurement being undertaken, may be included in RFP)

- 8. Fall clause. The following fall clause will form part of the contract placed on successful Bidder:-
  - (a) The price charged for the stores supplied under the contract by the Seller shall in no event exceed the lowest prices at which the Seller sells the stores or offer to sell stores of identical description to any persons/ Organisation including the purchaser or any department of the Central government or any Department of state government or any statutory undertaking the central or state government as the case may be during the period till performance of all supply Orders placed during the currency of the rate contract is completed.

- (b) If at any time, during the said period the Seller reduces the sale price, sells or offer to sell such stores to any person / organisation including the Buyer or any Deptt, of central Govt. or any Department of the State Government or any Statutory undertaking of the Central or state Government as the case may be at a price lower than the price chargeable under the contract, he shall forthwith notify such reduction or sale or offer of sale to the Director General of Supplies & Disposals and the price payable under the contract for the stores of such reduction of sale or offer of the sale shall stand correspondingly reduced. The above stipulation will, however, not apply to:--
  - (i) Exports by the Seller.
  - (ii) Sale of goods as original equipment at price lower than lower than the prices charged for normal replacement.
  - (iii) Sale of goods such as drugs which have expiry dates.
  - (iv) Sale of goods at lower price on or after the date of completion of sale/placement of the order of goods by the authority concerned under the existing or previous Rate Contracts as also under any previous contracts entered into with the Central or State Govt. Depts, including their undertakings excluding joint sector companies and/or private parties and bodies.
- (c) The Seller shall furnish the following certificate to the Paying Authority along with each bill for payment for supplies made against the Rate contract "We certify that there has been no reduction in sale price of the stores of description identical to the stores supplied to the Government under the contract herein and such stores have not been offered/sold by me/us to any person/organisation including the purchaser or any department of Central Government or any Department of a state Government or any Statutory Undertaking of the Central or state Government as the case may be upto the date of bill/the date of completion of supplies against all supply orders placed during the currency of the Rate Contract at price lower than the price charged to the government under the contract except for quantity of stores categories under subclauses (a),(b) and (c) of sub-para (ii) above details of which are given below ......"

#### 9 Risk & Expense clause.

- (a) Should the stores not be delivered within the time or times specified in the contract documents, or if defective delivery is made in respect of the stores or any installment thereof, the Buyer shall after granting the Seller 45 days to cure the breach, be at liberty, without prejudice to the right to recover liquidated damages as a remedy for breach of contract, to declare the contract as cancelled either wholly or to the extent of such default.
- (b) Should the items not perform in accordance with the specifications / parameters provided by the SELLER, the BUYER shall be at liberty, without prejudice to any other remedies for breach of contract, to cancel the contract wholly or to the extent of such default.

- (c) In case of a material breach that was not remedied within 45 days, the BUYER shall, having given the right of first refusal to the SELLER be at liberty to purchase, manufacture, or procure from any other source as he thinks fit, other stores of the same or similar description to make good:-
  - (i) Such default.
  - (ii) In the event of the contract being wholly determined the balance of the stores remaining to be delivered there under.

#### 10. Force Majeure clause.

- (a) Neither party shall bear responsibility for the complete or partial non performance of any of its obligations (except for failure to pay any sum which has become due on account of receipt of goods under the provisions of the present contract), if the non performance results from such Force Majeure circumstances as Flood, Fire, Earth Quake and other acts of God as well as War, Military operation, blockade, Acts or Actions of State Authorities or any other circumstances beyond the parties control that have arisen after the conclusion of the present contract.
- (b) In such circumstances the time stipulated for the performance of an obligation under the present contract is extended correspondingly for the period of time of action of these circumstances and their consequences.
- (c) The party for which it becomes impossible to meet obligations under this contract due to Force Majeure conditions, is to notify in written form the other party of the beginning and cessation of the above circumstances immediately, but in any case not later than 10 (Ten) days from the moment of their beginning.
- (d) Certificate of a Chamber of Commerce (Commerce and Industry) or other competent authority or organization of the respective country shall be a sufficient proof of commencement and cessation of the above circumstances.
- (e) If the impossibility of complete or partial performance of an obligation lasts for more than 6 (six) months, either party hereto reserves the right to terminate the contract totally or partially upon giving prior written notice of 30 (thirty) days to the other party of the intention to terminate without any liability other than reimbursement on the terms provided in the agreement for the goods received.
- 11. **Specification**. The following Specification clause will form part of the contract placed on successful Bidder:-

## The Seller guarantees to meet the specifications as per appendix 'A' of RFP

- 12. <u>Genuineness Certificate</u>. Bidders should be in position of Genuineness Certificate and are to produce as and when requested by technical evaluation committee (TEC).
- 13. Earliest Acceptable Year of Manufacture. 2019 Quality / Life certificate will need to be enclosed with the Bill.

- 14. <u>Transportation</u>. The stores shall be delivered to the ultimate consignee, The Secretary INSCB for Delhi based teams and the Joint Secretaries of INSCC (V), INSCC (MB), INSCC (K) and Oi/C Navy teams by road only. The date of issue of the Bill of Lading shall be considered as the date of delivery. No part transportation of items would be permitted. In case it becomes inevitable to do so, the Seller shall not arrange part transportation without the express/prior written consent of the Buyer.
- 15. **Quality**. The quality of the stores delivered according to the present Contract shall correspond to the technical conditions and standards valid for the deliveries of the same stores or specifications enumerated as per RFP and shall also include therein modification to the stores suggested by the Buyer. Such modifications will be mutually agreed to. The Seller confirms that the stores to be supplied under this Contract shall be new i.e. not manufactured before (Year of Contract), and shall incorporate all the latest improvements and modifications.
- 16. **Quality Assurance**. The items should be of the latest manufacture, conforming to the current production standard and having 100% defined life at the time of delivery.
- 17. Inspection Authority. The Inspection will be carried out by the Secretary INSCB, New Delhi 110 021 and user Rep of Navy teams. The mode of Inspection will be Departmental Inspection / User Inspection / Joint Inspection.
- 18. <u>Pre-Dispatch Inspection</u>. The following Pre-dispatch Inspection clause will form part of the contract placed on successful Bidder -
  - (a) The Buyer's representatives will carry out Pre-Dispatch Inspection (PDI) of the items in order to check their compliance with specifications in accordance with its usual standard procedures. Upon successful completion of such PDI, the Seller and Buyer will issue and sign a Certificate of Conformity as per the specimen at Form DPM-21 (Available in MoD website and can be given on request.
  - (b) The Seller shall intimate the Buyer at least 45 days before the scheduled date of PDI. The Buyer will send his authorised representative(s) to attend the PDI.
  - (c) The list of Buyer's representatives together with their particulars must be communicated by the Buyer at least 07 days in advance to apply for the necessary authorisations and clearances to be granted.
  - (d) The Buyer reserves the right not to attend the PDI or to request for postponement of the beginning of the PDI by a maximum of fifteen (15) days from the date fixed for such PDI in order to allow his representative(s) to attend such tests, in which cases he shall inform in writing the Seller within 15 days before the date of the beginning of the PDI. Should the Buyer request for such postponement, liquidated damages, if any, shall not apply. In case the Buyer informs the Seller within the period mentioned hereinabove that he cannot attend the PDI or in case the Buyer does not come at the postponed date requested by him for performance of the PDI as mentioned above, the Seller shall be entitled to carry out said tests alone as scheduled. The Certificate of Conformity and the Acceptance Test Report will be signed by the Seller's QA representative alone and such documents bearing the sole signature of the Seller's QA representative

shall have the same value and effect as if they have been signed by both the parties. In case Buyer does not elect to attend the PDI, the Buyer shall intimate the Seller in writing that it does not wish to attend the PDI.

- (e) The Seller shall provide all reasonable facilities and assistance to the Buyer's representative for safety and convenience in the performance of their duties at the Seller's premises.
- (f) All costs associated with the stay of the Buyer's Representative(s) at the place of the PDI is to be carried out, including travel expenses, boarding and lodging, accommodation, daily expenses, shall be borne by the Buyer.
- (g) The Seller shall provide Acceptance Test Procedure to the Buyer's QA Agency within one month from the signing of the Contract.
- Claims. The following Claims clause will form part of the contract placed on successful Bidder –
  - (a) The claims may be presented either: (a) on quantity of the stores, where the quantity does not correspond to the quantity shown in the Packing List/Insufficiency in packing, or (b) on quality of the stores, where quality does not correspond to the quality mentioned in the contract.
  - (b) The quantity claims for deficiency of quantity shall be presented within 45 days of completion of JRI and acceptance of goods. The quantity claim shall be submitted to the Seller as per Form DPM-22 (Available in MoD website and can be given on request).
  - (c) The quality claims for defects or deficiencies in quality noticed during the JRI shall be presented within 45 days of completion of JRI and acceptance of goods. Quality claims shall be presented for defects or deficiencies in quality noticed during warranty period earliest but not later than 45 days after expiry of the guarantee period. The quality claims shall be submitted to the Seller as per Form DPM-23 (Available in MoD website and can be given on request).
  - (d) The description and quantity of the stores are to be furnished to the Seller along with concrete reasons for making the claims. Copies of all the justifying documents shall be enclosed to the presented claim. The Seller will settle the claims within 45 days from the date of the receipt of the claim at the Seller's office, subject to acceptance of the claim by the Seller. In case no response is received during this period the claim will be deemed to have been accepted.
  - (e) The Seller shall collect the defective or rejected goods from the location nominated by the Buyer and deliver the repaired or replaced goods at the same location under Seller's arrangement.
  - (f) Claims may also be settled by reduction of cost of goods under claim from bonds submitted by the Seller or payment of claim amount by Seller through demand draft drawn on an Indian Bank, in favour of Principal Controller/Controller of Defence Accounts concerned.

- (g) The quality claims will be raised solely by the Buyer and without any certification/countersignature by the Seller's representative stationed in India.
- 20. <u>Warranty</u>. The following Warranty will form part of the contract placed on the successful Bidder:-
  - Except as otherwise provided in the invitation tender, the Seller hereby declares that the goods, stores articles sold/supplied to the Buyer under this contract shall be of the best quality and workmanship and new in all respects and shall be strictly in accordance with the specification and particulars contained/mentioned in contract. The Seller hereby guarantees that the said goods/stores/articles would continue to conform to the description and quality aforesaid for a period mentioned in enclosure to the RFP from the date of delivery of the said goods stores/articles to the Buyer or from the date of shipment/dispatch from the Seller's works whichever is earlier and that notwithstanding the fact that the Buyer may have inspected and/or approved the said goods/stores/articles, if during the aforesaid period the said goods/stores/articles be discovered not to conform to the description and quality aforesaid not giving satisfactory performance or have deteriorated, and the decision of the Buyer in that behalf shall be final and binding on the Seller and the Buyer shall be entitled to call upon the Seller to rectify the goods/stores/articles or such portion thereof as is found to be defective by the Buyer within a reasonable period, or such specified period as may be allowed by the Buyer in his discretion on application made thereof by the Seller, and in such an event, the above period shall apply to the goods/stores/articles rectified from the date of rectification mentioned in warranty thereof, otherwise the Seller shall pay to the Buyer such compensation as may arise by reason of the breach of the warranty therein contained.
  - (b) The items supplied should carry a warranty of **06 months** from the date of acceptance of Supply Order or supply whichever is later.
- 21. <u>Product Support:</u> The following Product Support clause will form part of the contract placed on successful Bidder
  - (a) The Seller agrees to provide Product Support for the stores and consumables subcontracted from other agencies/ manufacturer by the Seller for a maximum period of 12 months including 06 Months of warranty period after the delivery of stores.
  - (b) Any improvement/modification/up gradation being undertaken by the Seller or their sub suppliers on the stores/equipment being purchased under the Contract will be communicated by the Seller to the Buyer and, if required by the Buyer, these will be carried out by the Seller at Buyer's cost.

(Paresh Sawhney)

Commodore

President INSCB (AOL)

### Part V - Evaluation Criteria & Price Bid issues

- Evaluation Criteria The broad guidelines for evaluation of Bids will be as follows:
  - (a) Only those Bids will be evaluated which are found to be fulfilling all the eligibility and qualifying requirements of the RFP, both technically and commercially.
  - (b) In respect of Two-Bid system, the technical Bids forwarded by the Bidders will be evaluated by the Buyer with reference to the technical characteristics of the equipment as mentioned in the RFP. The compliance of Technical Bids would be determined on the basis of the parameters specified in the RFP. The Price Bids of only those Bidders will be opened whose Technical Bids would clear the technical evaluation.
  - (c) The Lowest Bid will be decided upon the lowest basic price quoted by the particular Bidder as per the Price Format given at Para 2 below. All taxes and duties (including those for which exemption certificates are issued) quoted by the Bidders will be not be considered to determine the L-1.
  - The Bidders are required to spell out the rates of Customs duty, Excise (d) duty, GST etc in unambiguous terms; otherwise their offers will be loaded with the maximum rates of duties and taxes for the purpose of comparison of prices. If reimbursement of Customs duty / Excise Duty / GST is intended as extra, over the quoted prices, the Bidder must specifically say so. In the absence of any such stipulation it will be presumed that the prices quoted are firm and final and no claim on account of such duties will be entrained after the opening of tenders. If a Bidder chooses to quote a price inclusive of any duty and does not confirm inclusive of such duty so included is firm and final, he should clearly indicate the rate of such duty and quantum of excise duty included in the price. Failure to do so may result in ignoring of such offers summarily. If a Bidder is exempted from payment of Customs duty / Excise Duty / GST upto any value of supplies from them, they should clearly state that no excise duty will be charged by them up to the limit of exemption which they may have. If any concession is available in regard to rate/quantum of Customs duty / Excise Duty / GST, it should be brought out clearly. Stipulations like, excise duty was presently not applicable but the same will be charged if it becomes leviable later on, will not be accepted unless in such cases it is clearly stated by a Bidder that excise duty will not be charged by him even if the same becomes applicable later on. In respect of the Bidders who fail to comply with this requirement, their quoted prices shall be loaded with the quantum of excise duty which is normally applicable on the item in question for the purpose of comparing their prices with other Bidders. The same logic applies to Customs duty and GST also.
  - (e) If there is a discrepancy between the unit price and the total price that is obtained by multiplying the unit price and quantity, the unit price will prevail and the total price will be corrected. If there is a discrepancy between words and figures, the amount in words will prevail for calculation of price.
  - (f) Any other criteria as applicable to suit a particular case.

2. <u>Price Bid Format</u>: The Bidders are required to fill it up correctly with full details in the format appended below:-

<u>SI</u>	Description/ Item Name	Unit price	HSN Code	GST	Custom duty	Any other tax/duties	Over all unit price	Qty	Total price
(a)									
(b)									
(c)									

- (a) Is GST extra?
- (b) If yes, then mention the following:
  - (i) Total value on which GST is leviable:
  - (ii) Rate of GST:
  - (iii) Total value of GST leviable:
- (c) Custom duty will be paid in actuals on submission of proof of payment.
- (d) Octroi / Entry taxes:
- (e) Any other Taxes / Duties / Overheads / Other costs:
- (f) Grand Total:
- 3. All the financial details are to be mentioned in the price bid only. Anything mentioned in any other part of the submitted documents having a financial bearing will not be considered.

(Paresh Sawhney)

Commodore

President INSCB (AOL)

		TO	RPF II	SCB/1098/SPORTS KIT & GEARS/19		endix ' May '
	SCHEDULE 'A' TO INSCB	TENDER	NO IN:	SCB/1098/KIT & GEAR/19-20 DATED	May 19	
SL	Name of Discipline/ Items	Denom	Qty	Specification & Make / Model	Delivery Unit	BID YES / NO
	Vest & Short Athletics	Sets	252	Dry Fit Fabric Sweat absorbant Sublimation With Navy Logo		NO
2	Spikes for Sprinter (100 M & 200M Sprint)	Pairs	21	Made of Thin TPU Material on the Toe, An EVA heel Wedge,with 6 mm pyramid spikes and wrench and weight should not more than 175 gms, (Nike Zoom Superfly Model No 835996-017, Puma Evo Speed Netfit Sprint Model No 19043301 Or Equivalent if as quoted model is not available)		
•	Spikes for 400 Mtr Hurdle	Pairs	11	Made of Thin TPU Material on the Toe, An EVA heel Wedge, and wrench and spikes weight should not more than 135 gms Nike Zoom 400 Model No 19092, Adidas Adizero Model No Q34041 (Or Equivalent if as quoted model is not available)		
	Spikes for 110 Mtr Hurdle	Pairs	13	Made of Thin TPU Material on the Toe, An EVA heel Wedge, with 5 mm pyramid spikes and wrench and weight should not more than 175 gms Puma Evo Speed Netfit Sprint Model NO 19043301, Nike Superfly Elite Racing Spikes Model No. 835996-017 (Or Equivalent if as quoted model is not available)	27	a: 2
	Spikes for Long Jump	Pairs	9	Made of Synthetic leather upper with no sew-technology fore foot strap and zippered entry for snug secure fit and Hard density EVA midsole for durable cushioning and midsole drop 5.5 mm and heel 14.5 mm fore foot 9 mm and weight not more than 236 grams size of UK 8.5 (Nike Zoom L/J 4 Model No. 415339-104, Adidas Adizero L/J Modal No. B22484, Asics Cyber Jump London Track Shoes Model No. B0058C60U0 Or Equivalent if as quoted model is not available)		
	Spikes for Triple Jump and Pole Vault	Pairs	13	Made of Synthetic leather upper with no sew-technology fore foot strap and zippered entry for snug secure fit and Hard density EVA midsole for durable cushioning and midsole drop 4 mm and heel 14 mm fore foot 10 mm and weight not more than 230 grams size of UK 8.5 (Adidas Adizero T/J - P/V Spikes Model No. BB6679, Asics Pro Flame Spikes, Nike Zoom Triple Jump Elite Model No. 474132103 Or Equivalent if as quoted model is not available)		

1	High Jump Spikes	Pairs	9	Made of light wt Synthetic and textile		
		11		with mid foot strap Molded EVA		a Sa
				midsole designed for durable	-	
				cushioning and weight not more than	11	n <sup>a</sup> .
			- 54	260 grams size of UK 9 Nike Elite		
		=	y.	Unisex Track Spikes Model No.		
				806561-446, Adidas Adizero H/J		
				Model No CG3835 (Or Equivalent if		
				as quoted model is not available )		
8	Javelin Spikes	Pairs	7	Made of 100 % Synthetic sole and		
				solid rubber for out sole weight not		
				more than 340gms Nike Javelin 2		
				Model No 3105500, Asics Cyber		
				Javelin Model No. G207Y.0190 (Or		
				Equivalent if as quoted model is not		
9	Shoes for Hammer, Discus &	Pairs	27	available ) Made of Thin TPU Material on the	77	
×	Shout Put Throwers	l all s	21	Toe, An EVA heel Wedge, with 6mm		
	onout at implicio			pyramid spikes and wrench and		
				weight should not more than 175 gms		
-			100	NoG207G0192, Adidas Shotput 2		
				Model No CG 3840 & Asics Cyber		
				Shotput Model G207g0190 Or		
				Equivalent if as quoted model is not		
				available)		
10	X- Country Shoes	Pairs	19	Nike Zoom fly Style-880848-001		
				Offset- 10mm Weight- 248 Gms,		
				Adidas Adizero Adios3 Item No-		
				AQ24358-C BLACKFTWWHJ/		
				RAYRED, Nike air zoom Pegasus 35		
				iD (Or Equivalent if as quoted model is not available)		
11	Spikes Long and Middle Distance	Pairs	17	Made of Thin TPU Material on the		15
50,000	Men's Running	, and		Toe, An EVA heel Wedge, and		
				wrench and wheight should not more		
				than 135 gms Nike Zoom Rival D		
				10ASIN-B078LL6CV2, New Balance		
				WLD 5000 Item No LD 5000v6,		
				Puma Men's Evospeed star 5 (Or		
				Equivalent if as quoted model is not		
				available)		
12	Athletics Walking Shoes	Pairs	13	Made of Thin TPU Material on the		
				Toe, An EVA heel Wedge, and		
	lif .			wrench and weight should not more		
				than 135 gms Nike Zoom Rival D		
				10ASIN-078LL6CV2, New Balance		
				WLD 5000 Item No LD 5000v6,		
	a			Puma Men, Nike Zoom fly Style- 880848-001 Offset- 10mm Weight-		3.
	8			248 Gms, Adidas Adizero Adios 3		
				Item No- AQ24358-C	-	_
				BLACKFTWWHJ/ RAYRED, Nike air		
				zoom Pegasus 35 iD (Or Equivalent if		
	Aquatics			as quoted model is not available )		

13	Swimming Costume	Nos.	105	FINA Approved Swimming Costume Make model and Specification:- (i) Speedo Model - 811443C138 Speedo Fit Power Mesh Pro Aquashort (ii) Speedo Model - 810826B733 Speedo Fit Panel Aquashort (iii) Speedo Model - 811348C587 Chevron splice Aquashort (Or Equivalent if as quoted model is not available)	
14	Swimming Goggles	Nos.	56	FINA Approved Swimming GogglesMake model and Specification:- (i) Speedo Model - 8108973515 Speed Socket 2Mirror Goggle (ii) Speedo Model-8083387678OV Opal Mirror Goggle (iii) Speedo Model - 810896B572 Fastskin Speed Socket (Or Equivalent if as quoted model is not available)	
15	Water Polo Cap	Sets	2	Water Polo Cap FINA Approved Make Model Specification:- Finis Model - WP-625008 (Or Equivalent if as quoted model is not available)	
16	Swimming Cap	Nos.	56	Swimming Cap FINA Approved Make model and Specification:- (i) Material - high quality silicone (ii) Resistance and protection against chlorinated water. (iii) Anatomic shape avoiding wrinkles (aa) Speedo Model - 8082160003 (ab) Speedo Model - 82160001 (ac) Speedo Model - 8216A715 (Or Equivalent if as quoted model is not available)	
17	Water Polo Ball (Competition/Practice)	Nos	22	Water Polo Ball Competition FINA Approved Make :- Mikasa (Or Equivalent if as quoted model is not	 -
18	Jammer Male	Nos.	21	available )  Jammer Male FINA Approved Make Model Specification:- (i) Speedo Model - 809755C292 Fastskin LZR Racer X High Waist Jammer . (ii) Speedo Model - 809755B454 Fastskin LZR Racer X High Waist Jammer. (Or Equivalent if as quoted model is not available )	
19	Water Stretch Cord (Black & Grey)	Nos	10	Water Strech Cord FINA Approved as per Specification:- (i) Cord is 06 meters long and is stretchable up to 20 meters, the cord has an internal safety core that prevents it from tearing. one end is attached to waist belt and the other to the side of the pool. (ii) Resistance rating of 1.3 - 3.6 Kg Make of Mad Wave (Or Equivalent if as quoted model is not available)	

20	Hand Paddle (Medium & Large)	Pairs	15	FINA Approved Hand Paddle Make of (i) Arena Model - 95232  Specification:- Vartex Evalution Hand Paddle (ii) Finis Model - 1.05.145 Specification:- Agility	= =	
21	Power Paddle (Medium & Large)	Pairs	15	FINA Approved Power Paddle as per Specification:- Speedo Model - 873156B076 Specification:- (Or Equivalent if as quoted model is not available)		
22	Kick Board	Nos.	15	FINA Approved Kick Board as per Specification:- Finis Model - 1.05.035 Specification:- EVA Form Speed Model - 802762C953 Specification:- EVA (Ethyeene Vinye Acctate) Form (Or Equivalent if as quoted model is not available)		
23	Pull Bouys Swimming	Nos.	15	FINA Approved Pull Bouys as per Specification:- (i) Speedo Model - 801791C572- ONESZ Specification:- Material – soft durable EVA form and Shape – figure eight (ii) Arena Model – 95056 Specification:- Material - light weight soft foam Composition: 80% Polyethylene, 20% Polypropylene) and Shape - figure eight (iii) Finis Model 1.05.036 Specification:- Material – EVA form and Shape - figure eight (Or Equivalent if as quoted model is not available)		
24	Cardiocap for Snorkel	Nos.	15	FINA Approved Cardiocap for Snorkel as per Specification:- Finis Model – 1.05.022 Specification:- Cardio cap for snorkel (Or Equivalent if as quoted model is not available)	-	
25	Snorkel	Nos.	15	FINA Approved Snorkel as per Specification:- Finis Model – 1.05.009 Specification:- (i) Center mount snorkel (ii) Stability during multiple strokes (iii) Comfortable silicone mouth piece (iv) one way purge valve (v) Adjustable head bracket (Or Equivalent if as quoted model is not available)		
26	Swimming Fins (Sizes XL,2XL& 3XXL)	Pairs	15	FINA Approved Swimming Fins as per Specification:- Finis Model - 1.05.037  Specification:- (i) Long blades provides resistance in the water.  (ii) Buoyancy from the fins lifts leg to the surface reducing drag from the rest of the body.  (iii) Natural Rubber (Or Equivalent if as quoted model is not available)		
27	Microfiber Towel ( Divers)	Nos	13	Microfiber Towel (divers) (i) Size - 120 x 60 CM (ii) primary material microfiber (80:20) Or Equivalent if as quoted model is not available)		

28	Head Bracket for snorkel	Nos	15	FINA Approved Head Bracket for Snorkel as per Specification:- Finis Model - 1.05.017 (i) Head bracket for Finis snorkel (Model - 1.05.009) (ii) Silicone strap - Fits snugly around the head or Equivalent (Or Equivalent if as quoted model is not available)		
	Basket ball	1	+			
29	Vest & short Basketball	Sets	64	Dry Fit Fabric Sweat absorbant both		-
	Vost a short Basketball	Joels	04	side Sublimation with Navy Logo		
30	Basket ball Shoes	Pairs	32	Nike zoom KD11Mid top product code 6677075, Under Armour men blue torch low Item no.7590613, Nike hyperdunk 2018 flykit high top Item No. (1963140) (Or Equivalent if as quoted model is not available)	4	
31	Basket ball Rubber (Practice)	Nos.	50	Cosco Tournament / Nivia Top Grip		
32	Basket ball Leather	Nos.	50	Nivia Pro Touch		
33	Basket ball Rings (Net)	Nos.	10	Nylon Thread 5mm thickness (Rptor or Equivalent)		
	Best Physique					
34	Posing Costume	Nos.	60	Lycra Nylon Spandex material		
35	T-Shirt & Shorts	Sets	42	Dry Fit Fabric Sweat absorbant Sublimation with Navy Logo		
36	T- Shirts Round Neck	Nos	42	Dry Fit Fabric Sweat absorbant Sublimation with Navy Logo		
37	Body Colour	Bottle	60	Protan (Or Equivalent if as quoted model is not available )		
38	Gym Shoes	Prs	21	Asics Men's gel excite 6 / New Balance (Or Equivalent if as quoted model is not available)		
39	Gym Belt Made of Good Quality of Leather	Nos	21	Kobo/ Costox (Or Equivalent if as quoted model is not available )		
40	Weight lifting Strap	Nos	21	Harbinger/Koba/Eleco (Or Equivalent if as quoted model is not available)		
	Boxing					
41	Vest/Short Boxing	Sets	84	Dry Fit Fabric Sweat absorbant both side Sublimation with Navy Logo		
42	Boxing Punching Pad	Nos.	11	Adidas Punching Pad, Sting Boxing VI Perpro Focus Pads (Or Equivalent if as quoted model is not available)		
43	Boxing Punching Gloves 10 Oz for Competition	Pairs	12	Top Ten		
44	Gum Shields	Nos.	21	Customized Gum Shields		
45	Boxing Shoes	Pairs	21	Nike Machomai Mid Boxing Boot Black, Title Boxing II Light weight Mid length Boxing Shoes (Or Equivalent if as quoted model is not available)		
46	Boxing Bandage	Pairs	50	Tranz / Vats / RXN / USI/ Everlast (Or Equivalent if as quoted model is not available)		
47	Boxing Punching Gloves 16 Oz Practice	Pairs	21	Tranz / Vats / RXN / USI (Or Equivalent if as quoted model is not available)		
48	Boxing Punching Gloves 12 Oz Competition	Pairs	12	Top Ten		

49	Boxing Traning Head Guard, Maximum, Large and XL	Pairs	10	Tranz / Vats / RXN / USI (Or Equivalent if as quoted model is not available )		
	Cricket			The state of the s		
50	T-Shirt H/S & Pant (White)	Nos.	52	Dry Fit Fabric Sweat absorbant White without Sublimation with Navy Logo		
51	Shirts F/S & Pant (White)	Nos.	52	Dry Fit Fabric Sweat absorbant White without Sublimation With Navy Logo		
52	T-Shirt H/S& Pant (Coloured)	Sets	26	Dry Fit Fabric Sweat absorbant with Sublimation With Navy Logo		
53	Shoes Batting	Pairs	26	(i) New Balance Batting Shoes Model - 4030v3, (ii) Nike Batting Shoes Domain 2 Model - 844125-108 (iii) Puma Batting Shoes Mens Evo Speed 18.1 C Spikemen Cricket shoes Model - 07DGQML3L (Or Equivalent if as quoted model is not available)		
54	Shoes Bowling	Pairs	26	(i) New Balance Bowling Shoes Model No - 4040v4 (ii) Asics Bowling Shoes Model No- P125Y.0192 (iii) Adidas Bowling Shoes Model No- CM 7417 (Or Equivalent if as quoted model is not available)	3	
55	Sweater Woollen White & Coloured	Nos.	52	Sweater Woollen coloured Pure wool with Embroided Navy Logo		
56	Cricket Kit Bag	Nos.	16	Cricket Kit Bag (dimension 98cmx40cmx40cm) with bat compartment and Large main compartment with zip opening along with side pocket for footwear and personnel items )		
57	Cricket Helmet (Size - S, M & L)	Nos.	10	Cricket Helmet Shrey Model No H121 Master Class Air Titanium/ Cricket Helmet Masuri M-VSTTLB Vision serier test titanium Model No B074WWRW1G (Or Equivalent if as quoted model is not available)		42
58	Cricket ball for Practice & Match	Doz	20	SG Test (Red Test Ball)		
59	Leg Guard (Batting)	Pairs	10	Leg Guard (Batting) New Balance Model No. 7TC 860 P / SG Premium Quality (Or Equivalent if as quoted model is not available)		
60	W/K Gloves	Pairs	6	New Balance Model - TC 1260 Wicket Keeping Gloves, SG Super Keep Men size & SS Limited edition wicket keeping Gloves (Or Equivalent if as quoted model is not available)		
61	W/K Leg Guard	Pairs	6	New Balance Model - TC560 Keeping Pad, SG Super Test & Puma Model No - EVO1 (Or Equivalent if as quoted model is not available)	a	
62	Cricket ball (White)	Doz	6	SG Test White Cricket Ball		
63	Batting Gloves	Pairs	22	New Balance Batting Gloves Model No- DC 1080, BAS Batting Gloves Model No SS Pro & SS Model No. SS Super Test Batting Gloves (Or Equivalent if as quoted model is not available)		

64	Thigh Pad	Nos.	10	New Balance Model - TC 1260 & Moon Walker Thigh Guard Model - ENDOS Thigh Guard (Or Equivalent if as quoted model is not		
				available )		
65	Grip for Bats	Nos.	100	SG, SS		
66	Stumps & Bails (3 Nos. each set)	Sets	6	Full Size, Made of good quality wooden		
67	Cricket Bat Practice	Nos.	10	SS (SS Gladiator English Willow Cricket Bat) with cover		
68	Cricket Bat Match	Nos.	10	SS (SS Maximus willow Cricket Bat) with cover		
69	Sun Glass Oakley (polarized)	Nos	26	(i) Oakley Men's Half Jacket 2.0 XL Iridium Sport Sunglasses (ii) Oakley Men's Radar OO9211-07 Shield Sunglasses (iii) Oakley Mirrored Other Men Sunglasses (0OO934305695802) (Or Equivalent if as quoted model is not available)		27
70	Hat Cricket SG	Nos	26	SG or Equivalent Brand		
71	T-Shirt & Shorts Football (Jersey)	Sets	122	Dry Fit Fabric Sweat absorbant with Sublimation with Navy Logo and chest number should be printed on front & back		4)
72	Foot ball shoes	Pairs	61	Nike Superfly 6 CL/Adidas Nemeziz 17.3 FG (Or Equivalent if as quoted model is not available)		
73	Shin Pad (Guard) Football	Pairs	32	Nike / Addidas(Or Equivalent if as quoted model is not available )		
74	Hand Gloves(GK)	Pairs	12	Nivia Hand Gloves (Or Equivalent if as quoted model is not available )		
75	T-Shirt & Short (GK)	Nos.	12	Dry Fit Fabric Sweat absorbant with Sublimation with Navy Logo and chest number should be printed on front & back		
76	Foot ball	Nos.	100	Nivia Shining Star or Equivalent		
77	Goal Post Net	Pairs	4	Good quality Nylon approved by FIFA		
78	Football Stocking	Pairs	122	Good quality Nylon Football Stocking		
	GOLF					
79	Golf Pant	Nos.	19	Callaway Dry Fit Article No. CGBS7076, Footjoy Article No 92203Adidas(Or Equivalent if as quoted model is not available)	8	
80	Golf Shoes	Pairs	19	Footjoy GSL PRO Article No 534965 , Callaway Aerosport BOA (Or Equivalent if as quoted model is not available)	1 12	
81	Hand Gloves (Golf)	Nos.	38	Foot Joy all weather, Callaway Fusion Pro		
82	Golf T-Shirt	Nos.	19	Foot Joy Article 31067 and Callaway Art No CGKS5072 with Navy Logo (Or Equivalent if as quoted model is not available)		- 4
83	Golf Ball Competition	Nos.	1000	Titliest PROVIX		-
84	Vest (Costume)	Sets	64	Dry Fit Fabric Sweat absorbant		
				Sublimation With Navy Logo		
85 86	Gymnastic Pant Aerobic Shoes	Nos. Pairs	64 18	Lycra Nylon Spandex material Asics Men's gel excite 6 / New Balance (Or Equivalent if as quoted		
				Balance (Or Equivalent if as quoted model is not available )		

87	Gymnastic Shoes	Pairs	34	IWA Gymnastic Shoes (Or Equivalent if as quoted item is not available )		
88	Gymnastic Grip for High Bar & Rings	pairs	34	Ri Sports / Christan Moori		
89	Swimming Costume For Gymnastic team Hockey	Nos.	34	Lycra Nylon Spandex material		
90	T Shirt / Shorts (Hockey)	Sets	122	Dry Fit Fabric Sweat absorbant with Sublimation with navy Logo and chest number should be printed on front & back		
91	Hockey Shoes (Turf) Training /Competition	Pairs	64	Adidas Flex Cloud Hockey shoes, Adidas Adipower Hockey Shoes, Asics Men Gel Hockey Shoes Neo 4 (Or Equivalent if as quoted model is not available)		
92	Shin Pad (Hockey)	Pairs	32	SNS Anatommic Fit Shin Guard, Adidas Lux Shin Guard & Osaka C Pro Shin Guard (Or Equivalent if as quoted model is not available)		
93	Hockey Stick (Graphite Senior)	Nos.	64	SNS Fully Composite Hockey Stick, Adidas DF/LX/VX Carbon Stick & Osaka Hockey Stick (Or Equivalent if as quoted model is not available)	ES	
94	Hockey Stick Bag (for 05 Stick and match kit)	Nos.	32	SNS Pro Tour Jumbo Hockey Stick Bag, Adidas Hockey Stick Bag for 5 Stick & Osaka SP Large Hockey Stick Bag) (Or Equivalent if as quoted model is not available)		-
95	Hockey Ball Turf (Training/Competition)	Nos.	80	Kokabura Elite		
96	Hockey Ball Turf (Traning)	Nos.	120	TK/Alfa/Vampire (Or Equivalent if as quoted items is not available)		
97	Finger Guard (Hockey)	Nos.	32	SNC Ultra Lite Glove, Gryphon G Mitt Pro 2018 Glove LH & Adidas Knuckle Guard 2018		
98	T-Shirt & Shorts Hockey (GK)	Sets	12	Dry Fit Fabric Sweat absorbant with Sublimation with navy Logo and chest number should be printed on front & back		
99	Hockey Stocking	Pairs	122	TK, Shiv Naresh, SNS Navy Blue and White Combination (Or Equivalent if as quoted Items is not available)	.7	
	Hockey GK Pad	Pairs	2	OBO HI Rebound		
	Hockey GK Gloves	Pairs	2	OBO HI Rebound		
	Gum Shield (Hockey) Shoes for Hockey GK	Nos Prs	8	Customized Gum Shields Asics Typhoon, Nike Hyper Venom (Or Equivalent if as quoted Model is not available)	,4	
104	Kicker for Hockey Goalkeeper	Pairs	2	OBO HI Rebound		
105	Goalkeeper Kit Bag	Nos	4	Grypbon/SNS/OBO (Or Equivalent if as quoted model is not available )		
	Handball					
106	T-Shirt & Shorts (Handball)	Sets	106	Dry Fit Fabric Sweat absorbant with Sublimation with navy Logo and chest number should be printed on front & back	F	

20						
107	Handball Shoes Practice	Pairs	28	Handball Shoes Practice Asics Model No. T805N GT-20006, Asics Model - T800N GEL Nimbus 20, Asics Model - T749N Men Gel Kayano 24 Running Shoes (Or Equivalent if as quoted model is not available)		
108	Handball Shoes Competition	Pairs	19	Handball Shoes Competition Adidas Stabil Optifit Shoes, Asics Model - T805N GT-20006, Asics Model - T800N GEL Nimbus 20 (Or Equivalent if as quoted model is not	3 1	
109	Upper & Lower Handball (GK)	Sets	12	available )  Dry Fit Fabric Sweat absorbant with Sublimation with navy Logo and chest number should be printed on front & back	(a)	
110	Handball	Nos.	90	Handball Nivia Model No 371 (Or Equivalent if as quoted model is not available)		
111	Inner (Skin) Tight Upper Full Sleeves	Nos	53	SG Specification:- ICON Skin - Fit (Black)	20	
112	Goal post net for Handball (Black)	Set	4	Rapter Nylon 3-5 mm thickness Made with high quality Nylon material and colour of net black		2/2
	Kayaking & Canoeing					
	Kayaking & Canoeing Tighty (Upper Full Sleeves & Lower)	Sets	51	Lycra Nylon Spantex material		
114	Shoes Waterproof for Kayaking & Canoeing  Rowing	Pairs	51	Adidas/Nike/ Asics(Or Equivalent if as quoted model is not available )		
115	Rowing Tighty (Upper Full Sleeves & Lower)	Sets	21	Lycra Nylon Spantex material		
116		Sets	21	Adidas/Nike/ Asics(Or Equivalent if as quoted model is not available )		
	Kabaddi					
117	T-Shirt & Shorts Kabaddi	Sets	82	Dry Fit Fabric Sweat absorbant with Sublimation with navy Logo and chest number should be printed on front & back		8 6
118	Kabaddi Mat Shoes	Pairs	41	Tranz Kabaddi Shoes		
	Squash					
119	T Shirt / Short (Squash)	Sets	18	Dry Fit Fabric Sweat absorbant Sublimation with Navy Logo		
120	Squash Shoes	Pairs	9	Salming Hawk Men's (210595), Head Revolt Indoor (273707), Asics Gel Fast ball 3 (Or Equivalent if as quoted model is not available)		
121	Squash Racket with Guts	Nos.	9	Head Graphene Touch Speed 120 Slimboy (211047), Head Graphene Redical 120 SB (210058), Technifibre Carboflex 125 x Speed		
122	Guts (Roll 200 Mtr each) Squash	Roll	5	Techni Fibre 305 Green (Or Equivalent if as quoted model is not available)		
123	Squash Racket Bag (Three Zipper x three large size compartment)	Nos.	5	Head Tour Team Travel BagWilson tour (K) factor three main zipper, Head tour three main zipper/Prince three main zipper/ Techni Fibre Tour VO2 Max 3 Zipper Dimension: 79 cmx33cmx16cm (Or Equivalent if as quoted model is not available)		* . ,
124	Squash Racket Replacement . Grip	Nos.	100	Karakal Replacement Grip / Head Replacement Grip / Technifibre		

125	Back Pack Squash Specific (minimum two large main compartment + compartment for shoes)	Nos	5	Head Tour Team Backpack/Technifibre Air Endurance Backpack bag (587996) (Or Equivalent if as quoted model is not	T.	5
126	Squash ball	Doz	25	available )  Dunlop Double Dot for Advance player (WISPA) (Or Equivalent if as quoted model is not available )		
	Tennis					
127	T-Shirt & Shorts (Tennis)	Sets	46	Dry Fit Fabric Sweat absorbant		
100			- 00	Sublimation with Navy Logo		
128	Tennis Racket with guts	Nos.	23	(i) Head Youtek Graphene Xt Speed MP (300 gm),(ii) Babalat Aero Pro Drive GT 2013 (300 gm),(iii)Wilson Blade 98 (Or Equivalent if as quoted model is not available)		
129	Strings (200 mtr) for Tennis	Roll	24	(i) Salinco Tour Bite, (ii) Outlast, (iii) Ram Blast		
130	Over Grip for Tennis Racket	Nos.	100	(i) Wilson Pro over grip Sensation, (ii) Babolat Pro Tour over grip, (iii) Salinco High Performanance over grip (Or Equivalent if as quoted model is not available)		
131	Tennis Racket Bag (minimum three large main compartment + compartment for shoes)	Nos.	12	Head Redical 12 R Monster Combi Pack/ Head Speed 12 PR Monster Combi Pack / Babalat Aero Pro Tou 12 Pack Bag (Or Equivalent if as quoted model is not available)	-	*
132	Tennis Shoes	Prs	23	Nike / Asics Gel Game - 5 / Adidas Barricade 6 (Or Equivalent if as quoted model is not available )		
133	Back Pack Tennis Specific	Nos	12	Wilson, Head, Babolat, Prince, Technifibre (Or Equivalent if as quoted model is not available)		
134	Tennis Ball	Doz	90	Wilson/Wilson Australian Open/Wilson US Open		
	Volleyball					
135	T Shirt/ Shorts (Volleyball)	Sets	88	Dry Fit Fabric Sweat absorbant with Sublimation with Navy Logo and chest number should be printed on front & back		
136	Volley Ball Shoes	Pairs	44	Mizuno Mens Wave Lightning Z2 Model No V602B44, Asics Mens Gel Rocket 7 Indoor Shoes Model No. B01BUKI8TZE (Or Equivalent if as quoted model is not available)		
137	Volley Ball Practice	Nos.	80	Cosco Supper Volley Genuine Father Pasted 18B Size 4 Spartan Super Volley (Or Equivalent if as quoted model is not available)		80
138	Volley Ball Net	Nos.	6	Cosco Volleyball Net Nylon ASIN.B01MSSJ1K8, Vixon Volleyball Net Tournament Nylon ASIN. B01MT35Z08 (Or Equivalent if as quoted model is not available)		
139	Volley ball Competition	Nos	10	Mikas MVA 200 Olympic Volley (Or Equivalent if as quoted model is not available)		
	Wrestling				10-	

140	Wrestling Shoes	Pairs	74	Asics Men's Split Second 9 Wrestling Shoes Red & Black, Nike Speed sweep VII Black/Black/White & Nike Speed Sweep VII - Navy /Red/White(Or Equivalent if as quoted model is not available)		-
141	Costume Lycra (Wrestling)	Nos.	140	Lycra Nylon Spandex material with Navy Logo		
	Weight. Lifting					
142	Costume Wt. Lifting	Nos.	68	Lycra Nylon Spandex material with Navy Logo		E.
143	Weight Lifting T-Shirt & Short	Prs	36	Dry Fit Fabric Sweat absorbant Sublimation With Navy Logo		
144	Weight Lifting Shoes	Prs	36	Addidas Adipower Weight Lifting Shoes, Nike Romaleos 3 Wt Lifting Shoes (Or Equivalent if as quoted model is not available)		
145	Knee Cap Eleiko Wt Lifting	Prs	36	Eleiko Knee Capt Wt Lifting (Or Equivalent if as quoted model is not available)		Sal .
146	Eleiko Olympic Wt Lifting Belt	Nos	19	Eleiko Wt Lifting Belt Genuine Leather (10 cm back & 5cm in Front for Wt Lifters	e a	
147	Wt. Lifting Wrist Wraps Eleiko	Paris	36	Wt. Lifting Wrist Wraps Eleiko (neoprene wrist wraps with velcro fasttening Length 200 mm & width 70mm for Competition and Traning Both		
148	Pulling Straps leather Eleiko for Wt Lifting	Paris	19	Pulling Straps in real leather Eleiko for Wt Lifting (length 600 mm & width 40 mm)		
	Common Kit Items					
149	Crepe Bandage (5 Mt X 10 Cm)	Roll	150	Johnson & Johnson or other reputed		
150	Crepe Bandage (5 Mtr X15 Cm )	Roll	150	Johnson & Johnson or other reputed brand		43
151	Sports kit Bag	Nos.	588	PVC bag polyester with heavy duty stainless steel Zipper, two side and one centre pocket, stainless steel buckles with printed Navy emblem, Size-L-24,B-12 and H-12'Tetron/Matti T/Q water proof trolley		
152	Practice Bag (Back Pack)	Nos	588	PVC bag polyester with heavy duty stainless steel Zipper, with printed Navy emblem, Tetron/Matti water proof		= #A)
153	Track Suit	Prs	660	Material not more than 70 Denier, 34 Filament Polyester yarn on the back and 70 denier, 100 filament microfiber on the front. shrink resistant and wrinkle free fabric high quality fabric, 100% water sweat absorbent, good quality zip and elastic, inter mesh lining with embroided Navy Logo. Zipper on Upper and Lower Pockets and Lower Bottom (make of Nike/Asics Or Equivalent if as quoted items is not available.)		

154	Wind Cheater	Prs	588	Quality of wind cheater fabric, water		Τ
				proof and 100% wind proof, good quality zip and elastic, inter mesh lining with embroided Navy Logo.  Zipper on Upper and Lower Pockets and Lower Bottom make of Asics / Nike or Equivalent if as quoted items is not available )		
155	Shoes Jogger	Prs	588	Shoes should be Light weight, weight not more than 10 Oz, sole with Gel firm grip with better air flow PVC/PU mesh with heavy duty material make of (New Balance /Skechers/Asics/ Mizuno or Equivalent if as quoted items is not available )	20	
156	Cap Jeans	Nos.	660	Jean drill cloth with Navy Logo embroideed with extendable Velcro strip at back		
157	Shorts / Half Pant	Nos.	1190	Material not more than 70 Denier, 34 Filament Polyester yarn on the back and 70 denier, 100 filament microfiber on the front. shrink resistant and wrinkle free fabric 100 %Sweat absorbant make of (Adidas/Nike /Reebok/Asics or Equivalent if as quoted model is not available)	,	
158	Socks Cotton	Pairs	2250	Addidas/Nike/Reebok White Cotton sports socks with thick base and round elastic band full size.		
159	Skinlet / Tighty	Nos.	200	Addidas/Nike/Reebok Lycra Nylon Spandex material (Or Equivalent if as quoted model is not available)		
160	T Shirts	Nos.	1190	Material not more than 70 Denier, 34 Filament Polyester yarn on the back and 70 denier, 100 filament microfiber on the front. shrink resistant and wrinkle free fabric high quality fabric, 100% water sweat absorbent with embroided with Navy Logo and make of Nike/ Asics or Equivalent if as quoted model is not available)		
161	Towel	Nos.	660	100% water absorbent soft fabric plain(Size-L-52", B-26") Bombay Dyeing (Or Equivalent if as quoted model is not available)		
162	Hand Towel	Nos	588	100% water absorbent soft fabric plain(Size-L-24", B-15") Tulip (Bombay Dyeing) (Or Equivalent if as quoted model is not available)		
163	Track Suit Lower for Shooters	Nos.	42	Micro Super Fine dry fit fabric sweat absorbant	29	
164	Vest&Short	Sets	100	Dry Fit Fabric Sweat absorbent Sublimation with Navy Logo		

155	Magnesium Powder	Kgs	80	Mg CO3 Magnesium Powder in 1Kgs pkts		
166	Sandals Floaters	Pairs	588	Puma/Adidas/Nike/Asics (Or Equivalent if as quoted model is not available)		
167	Blazers and Neck Tie	Nos	200	Cloth Raymond Safari /OCM Teri Wool, Steel button, Pocket Naval Crest with Golden Jery, Neck tie- Silk cloth with Naval crest weaved on the Tie.	er .	
168	Golf Ball	Nos.	6000	Callaway Super Soft, Titeliest DT Tru Soft (Or Equivalent if as quoted model is not available)		
169	Fast Pain Relief Spray (Heat based)	Nos.	300	Volini 40 Gms (Or Equivalent if as quoted model is not available )		
170	Fast Pain Relief (Tube)	Nos.	300	Volini 30 gms (Or Equivalent if as quoted model is not available )		
171	Fast Pain Relief Spray (Cold /Freeze based)	Nos.	100	Bio Freeze Hot and Cold Spray 118 MI (Or Equivalent if as quoted model is not available)		
172	Knee Cap(LP)	Nos.	200	LP		
173	Knee Cap	No	150	L P Open Petella Knee Support		7
	Anklet Support	Paris	150	Aerofit, LC and Doctor Choice (Or Equivalent if as quoted model is not available)		×
175	Wrist Support	Paris	150	Aerofit, LC and Doctor Choice (Or Equivalent if as quoted model is not available)		
176	Padded Knee Cap	Nos	100	Aerofit, LC and Doctor Choice (Or Equivalent if as quoted model is not available)		
177	Elector Plaster (Kinesiology Tape)	Nos	200	Elastoplast, Elasstic Fabric Roll (Kinesiology Tape) (2.5cmx 3cm) or Equivalent		
	Speed Training Items					
178	Anklet Weight (500 gm)	Pairs	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
179	Parachute for speed training	Nos	50	Vinex Parachute for Speed Trg (Parachute size 48 inch, carrying case, The Belt with Velcro is adjustable (Fits all sizes up to 48 inches Waist), Materiel – Polyester) Easily release into over speed and easily folds down to fit in the carrying bag.(Or Equivalent if as quoted Items is not available)		
180	Foam Roller (6 inches diameter) length 13 inches	Nos	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
181	Foam Roller (6 inches diameter) length 18 inches	Nos	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
182	Resistance Bands ( Set of Four Multy Colours as per thinckness )	Set	100	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
183	Loops Bands ( Set of Four Multy Colours as per thinckness )	Set	100	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		
184	Super Loops Bands ( Set of Four Multy Colours as per thinckness )	Set	100	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		
185	Resistant Tube (Set of Three Multy Colours as per thinckness)	Set	100	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		

186	Swiss Ball 75 cm	Nos	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)	25	1)
187	Swiss Ball 85 cm	Nos	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
188	Swiss Ball 95 cm	Nos	50	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
189	Egg Exerciser ( Set of Four Multy Colours as per thickness )	Set	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)	h =	
190	Massage Ball ( Set of Three Multy Colours as per thickness )	Set	100	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		12 10 30
191	Finger Gripper	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		
192	Hand Gripper	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		p .
193	Yoga Mat (6 feet x 2.5 feet)	Nos	200	Reebok, Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		2.
194	Ice Bag 1 Kg	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
195	Hot and Cold Therapy Gel Pack	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		
196	Exercise Ladder (Rope)	Nos	50	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )	25	
197	Plastic Cones 12 Inches Multi Colours	Nos	200	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
198	Disc Cones Multi Colours	Nos	200	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		
199	Speed Hurdles Height 6 Inches	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available)		
200	Speed Hurdles Height 12 Inches	Nos	100	Aerofit & Dr. Choice (Or Equivalent if as quoted Items is not available )		

# INTEGRATED HEADQUARTERS OF MINISTRY OF DEFENCE (NAVY) INDIAN NAVY SPORTS CONTROL BOARD, 7<sup>TH</sup> FLOOR, CHANAKYA BHAWAN CHANAKYA PURI, NEW DELHI – 110 021

Tele: 011-26887485

FAX No.: 011-26887485

#### **TENDER NOTICE**

- 1. Sealed quotations in two bids, Technical and commercial are invited by the President Indian Navy Sports Control Board, on behalf of the Chief of the Naval Staff, for procurement of Sports Kit and gear for the Navy teams at an estimated cost of R/off (Rs. 2,74,32,004.00 (Rupees Two Crores Seventy Four Lakhs Thirty Two Thousand Four only) including GST for delivery at stipulated Naval Station / Units within 90 days of placing the supply order.
- 2. Tender documents containing detailed terms and conditions as well as eligibility criteria for the parties to respond, may be obtained from the office of Indian Navy Sports Control Board (INSCB) at the above address, between 0930 to 1700 hrs on any working day, from 03 Jun 19. The tender documents are also available on internet website <a href="www.indiannavy.nic.in">www.indiannavy.nic.in</a>. The tender money of Rs 500.00 ( Five Hundred only) is to be deposited by the vendors through a crossed demand draft/MRO drawn in favor of "Controller of Defence Accounts (Navy) Delhi" as follows:-
  - (a) At the time of collection by those who wish to collect by hand.
  - (b) At the time of opening of Technical Bids by those who download the Tender Enquiry from website.
- 3. Tenderers are to deposit refundable Earnest Money Deposit minimum Rs 10000/- or (EMD) @ 3% of quoted items along with their bids. The EMD may be submitted in favor of the Secretary, Indian Navy Sports Control Board in the form of an Account Payee Demand Draft, Fixed Deposit Receipt, Banker's Cheque or Bank Guarantee from any of the public sector banks or a private sector bank authorized to conduct government business as per Form DPM-16, along with Technical Bids. The EMD is to remain valid as mentioned in Part I Para 15 of Tender Enquiry (RFP). The pre Bid meeting will be held at 1530 hrs on 18 Jun 19 in the office of the Secretary INSCB, New Delhi.
- 4. The tenders should be dropped in the box kept near the reception at Indian Navy Sports Control Board by 1330 h on 25 Jun 19. The Technical bids will be opened at 1400 h on 25 Jun 19 in the office of the Secretary, Indian Navy Sports Control Board, 7<sup>th</sup> Floor, Chanakya Bhawan, Chanakya Puri, New Delhi 21. The samples evaluation would be undertaken by the TEC as mentioned in Part –I Para 7 (C) of Tender Enquiry (RFP) at Chanakya Bagh, Chanakya Puri New Delhi 110021. Tenders found incomplete or received late will be rejected and no further correspondence on this matter will be entertained. The Department reserves the right to reject any or all tenders without assigning any reason thereof.